Author's response to reviews

Title: Differential associations of central and brachial blood pressure with carotid atherosclerosis and microvascular complications in patients with Type 2 diabetes

Authors:

- Chan-Hee Jung (chanh@schmc.ac.kr)
- Sang-Hee Jung (sangkerist@yahoo.co.kr)
- Kyu-Jin Kim (88014@schmc.ac.kr)
- Bo-Yeon Kim (byby815@schmc.ac.kr)
- Chul-Hee Kim (chkimem@schmc.ac.kr)
- Sung-Koo Kang (dmkang7003@schmc.ac.kr)
- Ji-Oh Mok (hanna@schmc.ac.kr)

Version: 2
Date: 12 December 2013

Author's response to reviews: see over
Dear Editors,

Thank you for your great efforts for *Cardiovascular Disorders*

We wrote ‘Acknowledgements’ section in the manuscript as you comment.

We now submit an original article entitled “Differential associations of central and brachial blood pressure with carotid atherosclerosis and microvascular complications in patients with Type 2 diabetes” which I wish to publish in the *Cardiovascular Disorders*

I declare that;

1) There has been no duplicate publication or submission elsewhere of any part of the work in any language, other than as an abstract.
2) The all co-authors have read the manuscript and approved its submission to, *Cardiovascular Disorders*

This paper has been written according to the guidelines provided by your journal and received assistance from a translation service.

I would appreciate very much if you could review it for publication in your journal.

I thank you in advance for your concern.

Sincerely yours,

Ji-Oh Mok, MD, PhD
Division of Endocrinology and Metabolism, Department of Internal Medicine, Soonchunhyang University School of Medicine,
#108 Jung-Dong, Wonmi-Ku, Bucheon, 110-746, Kyunggi-Do, Korea
Tel : 82-32-621-5156
Fax : 82-32-621-5016
E-mail : hanna@schmc.ac.kr